



# **Myoelectric Control Systems for Upper Limb Wearable Robotic Exoskeletons and Exosuits—A Systematic Review**

Jirui Fu <sup>1</sup>, Renoa Choudhury <sup>1</sup>, Saba M. Hosseini <sup>2</sup>, Rylan Simpson <sup>2</sup> and Joon-Hyuk Park <sup>1,\*</sup>

- <sup>1</sup> Department of Mechanical and Aerospace Engineering, University of Central Florida, Orlando, FL 32816, USA
- <sup>2</sup> Department of Electrical and Computer Engineering, University of Central Florida, Orlando, FL 32816, USA
- Correspondence: joonpark@ucf.edu

Abstract: In recent years, myoelectric control systems have emerged for upper limb wearable robotic exoskeletons to provide movement assistance and/or to restore motor functions in people with motor disabilities and to augment human performance in able-bodied individuals. In myoelectric control, electromyographic (EMG) signals from muscles are utilized to implement control strategies in exoskeletons and exosuits, improving adaptability and human–robot interactions during various motion tasks. This paper reviews the state-of-the-art myoelectric control systems designed for upper-limb wearable robotic exoskeletons and exosuits, and highlights the key focus areas for future research directions. Here, different modalities of existing myoelectric control systems were described in detail, and their advantages and disadvantages were summarized. Furthermore, key design aspects (i.e., supported degrees of freedom, portability, and intended application scenario) and the type of experiments conducted to validate the efficacy of the proposed myoelectric controllers were also discussed. Finally, the challenges and limitations of current myoelectric control systems were analyzed, and future research directions were suggested.

**Keywords:** myoelectric control; EMG-based control system; upper limb wearable robot; upper limb exoskeleton; upper limb exosuit; biomechanical model; pattern recognition; machine learning; reinforcement learning

## 1. Introduction

The demand for wearable robotic exoskeletons, more specifically upper limb rigid wearable robotic exoskeletons and soft wearable robotic exosuits, has substantially grown over the past few decades due to their promising applications across industry, and the medical and military sectors. The exoskeletons contain rigid links and joints attached to the user's body, while exosuits use soft, flexible materials (such as fabric or soft polymer) to interface with the user's body [1]. The intended application scenarios for upper limb exoskeletons and exosuits include: (i) power augmentation to enhance physical performance or the capabilities of able-bodied individuals during strenuous or repetitive physical tasks, (ii) assisting individuals with disabilities in performing activities of daily living (ADLs), and (iii) the rehabilitation of patients with neuromuscular disorders through therapeutic exercises [2].

The myoelectric control systems of upper limb exoskeletons are based on surface electromyography (EMG) signals, which are the electric potentials directly measured from skeletal muscles and that are generated from motor unit activation [3]. The generation of EMG signals is controlled by the human brain through motion intention, and is regulated by motor neurons in the spinal cord (Figure 1), which offers a means for detecting the human motion intention before initiating a motion [4]. Compared to other control systems, the critical advantage of myoelectric control is its timely detection of the user's motion intention leveraging electromechanical delay; the onset of motion intention can



Citation: Fu, J.; Choudhury, R.; Hosseini, S.M. ; Simpson, R.; Park, J.-H. Myoelectric Control Systems for Upper Limb Wearable Robotic Exoskeletons and Exosuits—A Systematic Review. *Sensors* 2022, 22, 8134. https://doi.org/10.3390/ s22218134

Academic Editor: Michael E. Hahn

Received: 13 September 2022 Accepted: 21 October 2022 Published: 24 October 2022

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). be detected about 50–100 ms earlier than the physical motion [5,6], giving time to control the upper limb wearable robotic exoskeletons to allow for more adaptive and intelligent human-robot interactions [7].



Figure 1. The generation of electromyography signal corresponding to the muscle contraction.

The early stages of myoelectric control modalities for upper limb exoskeletons were based on on-off/finite state control and proportional control [8,9]. Although they are fast, reliable, and applicable for real-time operation, the number of movements generated by these controllers is limited [10]. Therefore, they have been mostly utilized in supporting simple upper limb functions such as elbow flexion/extension or power grip. Recent advancements in machine learning (ML) and deep learning algorithms have made it possible to decipher more complex movements over multiple DOFs (such as shoulder motions) by extracting useful features from s-EMG signals [11]. Although they have shown promising results in supporting complex, multi-DOF upper limb motions, their development is still in the early phase.

Given the growing demand for exoskeletons, the number of publications on myoelectric control systems for upper limb exoskeletons and exosuits has rapidly increased over the past decade. It is therefore imperative to understand the latest trends and developments of myoelectric control systems for upper limb exoskeletons and exosuits, and to analyze their advantages and limitations in order to pinpoint the future directions for advancing this field. However, to the best of our knowledge, there has not been any comprehensive review that specifically focuses on the myoelectric control systems of upper limb exoskeletons and exosuits. Therefore, this review provides a comprehensive overview that will serve as a guidance for researchers and developers working in this field to address the challenges and limitations of existing myoelectric control systems, which in turn will enhance the utility and practicality of this technology and boost its real-world transition.

This review includes articles published in IEEE Xplore, Web of Science, and PubMed. Because the search result indicates that 194 out of 228 research articles were published between 2011 and 2021, this review will only focus on the research articles published during this duration. This review particularly focuses on the state-of-the-art myoelectric control systems designed for upper limb exoskeletons successfully implemented on a physical platform, i.e., exoskeleton/exosuit prototypes, and validated experimentally. The rest of the paper is organized as follows: Section 2 describes the methods used in conducting the literature review (e.g., eligibility criteria, search query, and screening). Section 3 summarizes the results of crucial design aspects, different myoelectric control modalities, and experimental validation methods found in the literature. Section 4 discusses existing challenges and outlines future research directions, and Section 5 provides the conclusion.

#### 2. Methods

The method used to search and to identify key literature pertaining to myoelectric control systems on upper limb exoskeletons are described in this section. This process was guided by the PRISMA method [12] for conducting a systematic review of the literature, based on specific search criteria and objectives.

#### 2.1. Eligibility Criteria

The eligibility criteria of this survey include articles: (i) written in English, (ii) published between 2011 and 2021, (iii) using s-EMG signals as inputs to control the upper limb wearable robotic exoskeleton, and (iv) implementing the controller on a physical model and conducting an experimental validation on human subjects. The time frame selected for this survey (criteria (ii)) was to capture the most recent advancements in myoelectric control systems for upper limb wearable robotic exoskeletons within the past 11 years. In addition, criterion (iv) was set to preclude studies in which myoelectric controllers were realized only in computational models or through simulation, since they cannot capture some uncertainties and nonlinearities present in the human–robot interface in the real world [13,14].

#### 2.2. Keywords and Search Query

The research articles reviewed in this paper were collected from three databases: IEEE Xplore, Web of Science, and PubMed, which are the database for the world's highest quality technical literature in engineering and technology (IEEE Xplore), the world's oldest, most widely used, and authoritative database of research publications and citations (Web of Science), and the world's largest free MEDLINE database for life sciences and biomedical literature (PubMed).

All three databases accepted similar search queries and keywords with small variations (Table 1). The keywords used were: (i) Body segments: "Upper Limb", "Upper Body", "Arm", "Shoulder", "Elbow", and "Forearm"; (ii) Types of Exoskeletons: "exoskeleton", and "exosuit"; (iii) Electromyography signal: "Electromyography", "Myoelectric", "Myoelectric control", "Surface Electromyography", "EMG", and "EMG".

Database	Search Query				
PubMed	(((myoelectric OR electromyography OR EMG) AND (upper limb OR elbow OR shoul-der OR hand OR finger OR wrist)) AND (exosuit OR exoskeleton)) AND (("2011/01/01" [Date—Publication] : "2021/12/31" [Date—Publication]))				
Web of Science	(((TS = (upper limb OR upper body OR wrist OR elbow OR shoulder OR finger OR hand)) AND (TS = (electromyography OR emg OR semg OR surface electromyography OR myoelectric)) AND (TS = (exoskeleton OR exosuit)) AND PY = (2011–2021)) NOT TS = (Passive)) NOT TS = (Lower Limb OR Hip OR Knee)				
IEEE Xplore	("All Metadata": upper limb OR "All Metadata": upper body OR "All Metadata": elbow OR "All Metadata": shoulder OR "All Metadata": wrist OR "All Metadata": hand OR "All Metadata": finger) AND ("All Metadata": electromyography OR "All Metadata": emg OR "All Metadata": EMG OR "All Metadata": myoelectric) AND ("All Metadata": exosuit OR "All Metadata": exoskeleton)				

**Table 1.** Description of search query used in each database.

The search query resulted in 180 articles, among which 90 were counted twice, as they appeared from more than one database. In addition, based on criteria (iv), i.e., the implementation and validation of the proposed myoelectric control system on a physical model, 30 articles were excluded. Finally, 60 articles were selected and reviewed in this survey. Figure 2 illustrates the https://www.overleaf.com/project, (accessed on 12 September 2022) process of searching, screening, filtering, and the selection of articles to be included [12].



Figure 2. Process of searching, screening, filtering, and the selection of research articles to be included.

#### 3. Results

This result section contains the description and analysis of the included research articles. A narrative style is adopted to present our results. The results are presented based on four aspects: (i) a general description of information sources, (ii) different types of myoelectric control systems, (iii) key design characteristics of the device (i.e., DOF, portability, and intended application scenario), and (iv) types of experiments conducted to evaluate the efficacy of the proposed controller.

### 3.1. Overview of Information Sources

A summary of the relevant characteristics of the 60 articles included in this literature review is organized in Table 2, which includes: bibliographic citation and information of authors of the article (Ref and Author), publication year of the article (Year), upper limb segments assisted by the exoskeleton (Body Segment), types of myoelectric control method (Control Method), portability of the device (Device Portability), supported degree of freedom (DoF), intended application scenario (Application of Device), and number and health condition of subjects in experimental validation (Experimental Subject).

No.	<b>Ref and Author</b>	Year	Body Segment	<b>Control Method</b>	<b>Device Portability</b>	DoF	Application of Device	<b>Experimental Subject</b>
1	Ho et al. [15]	2011	Hand	Threshold	Portable	10	Rehabilitation	8 chronic stroke subjects
2	Lenzi et al. [9]	2012	Elbow	Proportional	Fixed	1	Augmentation and Assistance	10 healthy subjects
3	Gopura and Kiguchi [16]	2012	Shoulder, Elbow, Wrist	Neural Fuzzy	Fixed	6	Rehabilitation	1 healthy subject
4	Kiguchi and Hayashi [17]	2012	Shoulder, Elbow, Wrist	Neural Fuzzy	Fixed	7	Rehabilitation and Assistance	3 healthy subjects
5	Pang et al. [18]	2012	Hand	Machine Learning	Portable	1	Rehabilitation	3 healthy subjects
6	Delph et al. [19]	2013	Hand	Proportional	Portable	1	Rehabilitation	Not Specified
7	Loconsole et al. [20]	2013	Hand	Machine Learning	Portable	1	Rehabilitation	1 healthy subject
8	Su et al. [21]	2013	Elbow	Machine Learning	Fixed	3	Rehabilitation	1 healthy subject
9	Ngeo et al. [22]	2013	Hand	Machine Learning	Portable	3	Assistance	1 healthy subject
10	Ramos and Meggiolaro [23]	2014	Elbow	Model Base	Portable	2	Augmentation	1 healthy subject
11	Kawase et al. [24]	2014	Elbow, Wrist	Model Base	Portable	6	Rehabilitation	6 healthy subjects, 1 SCI patient
12	Loconsole et al. [25]	2014	Elbow	Model Base	Fixed	1	Rehabilitation	1 healthy subject
13	Li et al. [26]	2014	Elbow, Wrist	Machine Learning	Fixed	2	Assistance	5 healthy subjects
14	Li et al. [ [27]	2014	Elbow	Machine Learning	Portable	1	Rehabilitation	6 healthy subjects

**Table 2.** General summary of the research papers included in this review. Corresponding to Citation of document, Author Info., Publish Year, Included BodySegments, Type of Myoelectric Control System, Portability, DoF, Application of Exoskeleton, and Characteristics of Experimental Subject.

Table 2. Cont.

No.	<b>Ref and Author</b>	Year	Body Segment	Control Method	<b>Device Portability</b>	DoF	Application of Device	Experimental Subject
15	Kirchner et al. [28]	2014	Elbow, Wrist	Machine Learning	Portable	4	Rehabilitation	8 healthy subjects
16	Buongiorno et al. [29]	2015	Shoulder, Elbow	Model Base	Fixed	4	Rehabilitation	3 healthy subjects
17	Riener and Novak [30]	2015	Elbow	Threshold	Fixed	7	Rehabilitation	3 healthy subjects
18	Krasin et al. [31]	2015	Elbow	Threshold	Portable	1	Augmentation	Not specified
19	Leonardis et al. [32]	2015	Hand	Machine Learning	Portable	1	Rehabilitation	6 healthy subjects, 2 chronic stroke patients
20	Ullauri et al. [33]	2015	Elbow	Model Base	Fixed	1	Rehabilitation	2 healthy subjects
21	Triwiyanto et al. [34]	2016	Elbow	Proportional	Fixed	1	Rehabilitation	Not specified
22	Peternel et al. [35]	2016	Elbow	Model Base	Fixed	1	Rehabilitation and Assistance	8 healthy subjects
23	Accogli et al. [8]	2017	Wrist, Elbow	Machine Learning	Fixed	4	Assistance	1 healthy subject
24	Lu et al. [36]	2017	Hand	Machine Learning	Fixed	5	Rehabilitation	8 healthy subjects; 2 SCI subjects
25	Li et al. [37]	2017	Elbow	Model Base	Fixed	2	Assistance	1 healthy subject
26	Hosseini et al. [38]	2017	Elbow	Threshold	Portable	2	Assistance	1 healthy subject
27	Mghames et al. [39]	2017	Elbow	Proportional	Portable	1	Augmentation	1 healthy subject
28	Hamaya et al. [40]	2017	Elbow	Machine Learning	Fixed	1	Assistance	5 healthy subjects
29	Yun et al. [41]	2017	Hand	Machine Learning	Portable	8	Augmentation	2 SCI Patients

Table 2. Cont.

No.	Ref and Author	Year	Body Segment	Control Method	Device Portability	DoF	Application of Device	Experimental Subject
30	Irastorza-Landa et al. [42]	2017	Wrist	Machine Learning	Portable	7	Rehabilitation	8 healthy subjects
31	Lambelet et al. [43]	2017	Wrist	Proportional	Portable	1	Rehabilitation	1 healthy subject
32	Lince et al. [44]	2017	Hand	Proportional	Portable	4	Rehabilitation	8 healthy subjects
33	Copaci et al. [45]	2018	Elbow	Threshold	Portable	1	Rehabilitation	1 healthy subject
34	Zeng et al. [46]	2018	Hand	Machine Learning	Portable	6	Rehabilitation	25 healthy subjects
35	Buongiorno et al. [47]	2018	Elbow	Model Base	Fixed	4	Augmentation and Assistance	7 healthy subjects
36	Cisnal et al. [48]	2019	Hand	Proportional	Portable	10	Rehabilitation	Not Specified
37	Jana et al. [49]	2019	Hand	Machine Learning	Portable	2	Rehabilitation	1 healthy subject
38	Trigili et al. [50]	2019	Elbow, Shoulder	Machine Learning	Fixed	4	Augmentation	10 healthy subjects
39	Lei [51]	2019	Elbow, Wrist	Machine Learning	Portable	1	Rehabilitation	4 healthy subjects
40	Wu et al. [52]	2019	Elbow	Machine Learning	Portable	1	Rehabilitation and Assistance	5 healthy subjects
41	Xiao [53]	2019	Elbow	Machine Learning	Fixed	1	Rehabilitation	9 healthy subjects
42	Lu et al. [54]	2019	Hand	Machine Learning	Portable	5	Rehabilitation	12 SCI Patients
43	Secciani et al. [55]	2019	Hand	Machine Learning	Portable	4	Assistance	1 patient with hand impairment
44	Burns et al. [56]	2019	Hand	Machine Learning	Portable	10	Rehabilitation	5 healthy subjects
45	Lu et al. [57]	2019	Elbow	Machine Learning	Portable	1	Assistance	5 healthy subjects
46	Li et al. [58]	2020	Elbow	Machine Learning	Portable	1	Assistance	10 healthy subjects

Table 2. Cont.

No.	<b>Ref and Author</b>	Year	Body Segment	Control Method	Device Portability	DoF	Application of Device	Experimental Subject
47	Lotti et al. [59]	2020	Elbow	Model Base	Portable	1	Assistance	6 healthy subjects
48	Hosseini et al. [60]	2020	Elbow	Threshold	Portable	2	Assistance	4 healthy subjects
49	Da Silva et al. [61]	2020	Elbow	Proportional	Portable	2	Assistance	1 healthy subjects
50	Treussart et al. [62]	2020	Elbow	Proportional	Fixed	1	Augmentation	10 healthy subjects
51	Liu et al. [63]	2020	Elbow	Model Base	Fixed	1	Rehabilitation and Assistance	5 healthy subjects
52	McDonald et al. [64]	2020	Elbow	Machine Learning	Portable	4	Rehabilitation	10 healthy subjects, 1 SCI patient
53	Castiblanco et al. [65]	2021	Hand	Neural Fuzzy	Portable	4	Rehabilitation	4 stroke patients and 3 healthy subjects
54	Liu et al. [66]	2021	Elbow	Machine Learning	Portable	1	Rehabilitation	20 healthy subjects
55	Yang et al. [67]	2021	Elbow	Machine Learning	Portable	1	Rehabilitation	10 healthy subjects
56	Zhou et al. [68]	2021	Shoulder	Machine Learning	Fixed	5	Rehabilitation	18 healthy subjects
57	Cisnal et al. [69]	2021	Hand	Threshold	Portable	5	Rehabilitation	10 healthy subjects
58	Xiao et al. [70]	2021	Hand	Machine Learning	Fixed	7	Rehabilitation	9 healthy subjects
59	Liu et al. [71]	2021	Shoulder, Elbow	Proportional	Portable	2	Augmentation	3 healthy subjects
60	Xie et al. [72]	2021	Hand	Model Base	Portable	8	Rehabilitation	Not specified

In addition, we conducted a bibliometric analysis of the selected articles and plot (Figure 3) to present the network of relationships between the keywords of the myoelectric control system and upper limb wearable robotic exoskeletons associated with the documents examined. The bibliometric analysis reveals the evolution of the topics related to the myoelectric control system of upper limb exoskeletons over time, and their connections with other key terms.



**Figure 3.** The chronicle evolution of the topics related to the myoelectric control system of upper limb exoskeleton and interrelationship between each keywords. The map is generated with VOSViewer [73].

#### 3.2. Taxonomy of the Myoelectric Control System for Upper Limb Robotic Wearable Exoskeletons

The myoelectric control system is designed to detect the movement intention of the user by deciphering the EMG signals generated in skeletal muscles. Among the reviewed literature, different modalities of myoelectric control systems for upper limb exoskeletons were identified (as shown in Figure 4), which include: (i) Threshold-based control (12%, seven articles), (ii) Proportional Control (18%, ten articles), (iii) biomechanical model-based control (18%, ten articles), (iv) machine learning and deep learning-based control (47%, twenty-eight articles), and (v) neural-fuzzy control (5%, three articles). The ML and deep learning-based myoelectric control systems were further sub-categorized into (a) pattern recognition or classification-based control, (b) simultaneous or regression-based control, or (c) reinforcement learning-based control.



**Figure 4.** The percentage distribution of various types of myoelectric control systems among the included research literature, which includes biomechanic model-based, machine learning-based, proportional, threshold-based, and neural-fuzzy-based myoelectric control systems.

### 3.2.1. Threshold-Based Myoelectric Control

Threshold-based myoelectric control is a method that regulates specific control rules by comparing the amplitude [31] or statistical features [15] of EMG signals with certain thresholds to assist the user (as shown in Figure 5). The studies at [38,60] implemented threshold-based myoelectric control to assist the user in the flexion and extension of the forearm with different loads in hand. The proposed controller utilized a Double Threshold strategy based on EMG signals corresponding to Bicep Brachii (BB) and Triceps Brachii (TB). When the user tried to flex his/her arm with a load, the EMG signal corresponding to BB exceeded its threshold value (i.e., the amplitude of BB signal at the no-load condition) which activated the exoskeleton to support the user in the intended motion. Similarly, during forearm extension motion under load, the EMG signal corresponding to TB was greater than its threshold (i.e., the amplitude of TB signal at the no-load condition), which triggered the exoskeleton to assist the user during the performance of extension. Experiments in [38,60] validated the proposed control strategy under various load conditions with the same threshold values. It was demonstrated that the exoskeleton could successfully assist the user in supporting the motion by reducing his/her muscle effort, showing the adaptability of this strategy under different load conditions.



Figure 5. The conceptual block diagram for threshold-based (On-off/Finite State Machine) myoelectric control system.

To improve the accuracy of threshold-based control, [30,45] utilized sensor fusion methods are utilized to determine thresholds to modulate assistance. In [45], signals from

both EMG and force-sensing resistors (FSRs) were used to detect the user's motion intention. By measuring the contact force between the user and the robot, FSRs were utilized to detect the user's motion intention. When both EMG and FSR signals exceeded their thresholds, the assistance of a wearable robotic exoskeleton was engaged. In [45], the end-effector position (measured by IMU sensors), joint torques (estimated from force/torque sensors), and EMG signals were used in the sensor fusion method. The proposed control strategy in [30] was intended to assist patients with upper limb impairment in pointing his/her arm to the desired position; therefore, a fusion of end-effector position with joint torques and EMG signals enabled the user to track his/her desired trajectory more accurately. Same as [45], the conditions for controlling assistance in [30] were the end-effector position not being at the target position, and both joint torques and amplitudes of EMG signals being greater than their threshold values. Overall, threshold-based myoelectric control systems have the lowest computational cost. However, this control strategy can only support a limited number of movements at a time and is sequential in nature, unlike the continuous natural movement of the human upper limb.

#### 3.2.2. Proportional Myoelectric Control

Proportional myoelectric control utilizes a transfer function to continuously map the processed EMG signals to the dynamics of upper limbs (e.g., force, velocity, position, etc.), then uses it as a control input to assist the user in performing a specific movement (as shown in Figure 6). Whether the EMG signal is directly mapped to the desired joint angle or torque, this control scheme is divided into direct and indirect proportional control.



Figure 6. The conceptual block diagram for proportional/directed myoelectric control system.

Direct proportional control scales the processed EMG signal to the muscle effort because the amplitude of EMG signals infer the amount of muscle contraction [56,57] and estimate the generated joint torque [58]. Based on this notion, one can develop a direct myoelectric control system that maps the amplitude of EMG signals with a scaling factor to estimate the user's effort, which can then be used as a control input. The implementation of a direct proportional myoelectric control system for upper limb wearable robotic exoskeletons was first introduced by Lenzi et al. [9]. This work used a proportional gain as a scaling factor to estimate the user's effort using EMG signals. The estimated user's effort was then used as input to control the assistance of a 1-DOF elbow exoskeleton. The proportional gain was tuned manually with the control goal of the gravity compensation of the arm and the carried load, while performing repetitive arm motions. The experiment showed an average reduction of 31% in the user's muscular effort, with acceptable accuracy in maintaining the desired trajectory. In addition to [9,74], a similar approach has been implemented by other studies [19,43,44,48] on hand and wrist exoskeletons, and similar performances of controllers were found. However, in these studies, an additional low-level controller was required to regulate the actuator. In [34], the amplitude of EMG signals was scaled to the

duty cycle of a PWM controller, and the myoelectric control system was directly able to regulate the actuator. Compared to [9,19,43,44,48], the subsequent approach required less computational power, improving the exoskeleton's portability and useability.

Linear transfer functions were used to map the EMG signals in [9,19,43,44,48], but [62] used a nonlinear transfer function to represent the relationship between joint torque and EMG signal [75]. This nonlinear function was derived using experimental data; therefore, it is expected to provide more accurate joint torque estimation. Apart from using a nonlinear transfer function for joint torque estimation, the myoelectric control system in [62] also implemented a neural network model to estimate the direction of joint torque using the EMG signal. Other studies implementing indirect proportional control in the literature [39,61] utilized an admittance control system to map the EMG signal to the desired joint kinematics (i.e., joint angle or angular velocity). Compared to direct proportional myoelectric control systems, indirect myoelectric control systems have been shown to improve the trajectory tracking performance of the upper limb wearable robotic exoskeletons. Overall, proportional myoelectric control is easier to implement and shows robust performance in assisting the user in performing simple motion tasks such as elbow flexion/extension. The limitation is that it is not suitable for supporting complex, sophisticated multi-joint motion tasks in patients with upper limb impairment.

#### 3.2.3. Biomechanical Model-Based Myoelectric Control

A biomechanical model-based control system employs a musculoskeletal model to map the EMG signals to the desired control inputs, where the human musculoskeletal model is represented by a mechanical multi-body system actuated by muscle segments (as shown in Figure 7). Hill's Muscle Model was the only biomechanical model identified in the reviewed literature that was utilized in myoelectric control of upper limb exoskeletons.



Figure 7. The conceptual block diagram for bio-mechanics model-based myoelectric control system.

The survey results indicated that only a small portion of the reviewed articles (about 18%, 10 articles) used biomechanical model-based myoelectric control systems, and most were intended for power augmentation. The first implementation of Hill's Muscle Model in the myoelectric control system was presented by [76] to control a single DOF upper limb exoskeleton to assist the user in elbow flexion/extension. The control system used Hill's Muscle Model to estimate the moment generated by the Bicep Brachii and Triceps Brachii medial-head muscles as a primary input to the controller. The exoskeleton's kinematic and dynamic effects were examined experimentally on the assisted arm segments [77]. The results showed a reduction of around 86% in muscular activity and joint torque using the assistive system.

The authors in [59] also implemented a similar approach to develop an adaptive biomechanical model-based myoelectric control system for a one-DOF upper limb exosuit.

The purpose of the proposed myoelectric controller was to estimate the joint torque without using the inverse dynamic model. Instead, Hill's Muscle Model was implemented to map the EMG signal to the joint torque, and then the estimated joint torque was used as control inputs to a closed-loop admittance controller to detect motion intention and to compensate for the estimated joint torque. Their experimental results indicated that the proposed myoelectric control system had an average coefficient of determination ( $r^2$ ) 0.87 ± 0.04 (mean ± SD) across different human subjects, which demonstrated the robustness of the proposed system. Furthermore, for elbow motion assistance under various loads, the average  $r^2$  was 0.92 ± 0.03, which demonstrated the good adaptability of the system.

One limitation of Hill's Muscle Model is that there are situations when it cannot yield an accurate detection of the user's motion intention [78]. For example, there are cases where muscle activations occur without necessarily involving joint movement (e.g., isometric contraction) [79]. Therefore, a myoelectric control system using Hill's Muscle Model could misinterpret the user's motion intention, resulting in incorrect control commands. To address this limitation, Lotti et al. [59] introduced an additional encoder installed at the elbow joint to detect the kinematic velocity of elbow flexion/extension, then used them as another input signal for the controller. On the other hand, Liu et al. [63] introduced another biomechanical model-based myoelectric control system with the capability of movement onset detection, but only with the use of the EMG signal as input. The proposed myoelectric control system in [63] combined Hill's Muscle Model with a support vector machine (SVM) to control an upper limb exoskeleton for load-lifting assistance. The Hill's Muscle Model was used to map the EMG signal to the joint torque and it was used as the control command, similar to [59]. In addition, an SVM-based binary classifier was also incorporated for detecting the movement onset, and it activated the actuator once the movement onset was detected. Compared to [59], the integration of Hill's Muscle Model and an SVM-based classifier eliminates the delay in the control system while ensuring its accuracy. However, some limitations of the approaches in [59,63] include repetitive calibrations for different subjects and/or tasks, a long setup time, a high computational cost, and a lack of continuous control [80], and it requires many muscle segments for complex joints (e.g., shoulder joint) [81].

To reduce the setup time and to use fewer numbers of muscle segments for the shoulder joint muscle, ref. [23,29] used an optimization algorithm to tune the parameters in Hill's Muscle Model. Different from the conventional calibration approach that used the human's anatomical model [82], this work measured the arm's gravitational torque and derived the expression of muscle torque from Hill's Muscle Model, and then it used a Genetic Algorithm (GA) to optimize the parameters of the muscle model by minimizing the difference between the gravitational torque and muscle torque. Another study in [47] compared the performances of the GA and Linear Optimization Algorithms in tuning the parameters of Hill's Muscle Model. The results indicated that Hill's Muscle Model, tuned using a GA, showed a higher accuracy in estimating the joint torque ( $R^2 = 0.9$ for GA vs.  $R^2 = 0.8$  for Linear Optimization). Furthermore, compared to [59], which required one extra session to calibrate the muscle model, the GA optimizer in [23,29] only required two minutes to find the optimal parameters of the muscle model. Even though the work in [23,29] significantly reduced the calibration time compared to [59], the results still showed an average reduction of 67% in the user's muscular effort for the load lifting task, and an average increase of 8% in the  $R^2$  score after the GA optimization.

The research articles mentioned above use Hill's Muscle Model to estimate the joint torque; however, it can also be used in the myoelectric control system for other purposes. For example, ref. [37] developed a human-in-the-loop adaptive impedance control system for an elbow exoskeleton. The exoskeleton in [37] is designed for upper limb rehabilitation; therefore, both the predefined trajectory and the EMG signals were used as the control command. To adapt the control gains to the human muscle's impedance, this work used Hill's Muscle Model to calculate the muscle stiffness in real time, which is proportional to the joint torque from the measured EMG signals. The benefit of such adaptive impedance

control systems in the upper limb rehabilitation task is the accuracy of its output trajectory with respect to the desired trajectory (which is almost close to 100%).

Furthermore, ref. [35] used the EMG signal and Hill's Muscle Model as a bio-feedback of the myoelectric control system. Hill's Muscle Model was still used as a joint torque estimator in this work, similar to [29,47,59]. However, it used a predefined trajectory as a control command, and the goal was to minimize the biofeedback measured from the EMG signal. The minimization of biofeedback is targeted to reduce the error between the predefined trajectory and the actual trajectory.

As validated in many studies [24,29,35,37,47,59,63,72] the model-based myoelectric control systems are effective and versatile in controlling upper limb wearable robotic exoskeletons using EMG signals as the control input. Nonetheless, some limitations in the model-based myoelectric controls remain, such as a high computational cost, requiring repeated calibrations, and not being robust to unmodeled and/or external disturbances.

#### 3.2.4. Machine Learning-Based Myoelectric Control

The machine learning approach has been employed in upper limb exoskeletons to predict the movement pattern from EMG signals using various modalities. Such modalities of the myoelectric control system use the statistical features of the preprocessed EMG signals, such as root mean square (RMS), wave length (WL), and mean absolute value (MAV) [83] as inputs to different machine learning models, detecting the type of movement via classification model [21,49], measuring the arm's joint angle or joint torque via regression model [27,32], and minimizing muscle effort during assistance with reinforcement learning [40] (as shown in Figure 8).



**Figure 8.** The conceptual block diagram for deep learning/machine learning model-based myoelectric control system.

1. Pattern recognition or classification-based myoelectric control

In the myoelectric control system, the classification model can detect the type of movement corresponding to the input EMG signal. For example, ref. [49] implemented the SVM algorithm to predict whether the user's finger is in motion or not via the root mean square features of the EMG signal; the authors in [21] implemented a type of decision-tree algorithm named MCLPBoost to predict whether the elbow and wrist are performing downward or upward movements. In the abovementioned studies, only a single type of classification algorithm was used, and a comparison between different algorithms was not conducted. To compare the performance of different classification algorithms implemented on a hand exoskeleton, the authors in [84] compared the accuracy of four classification algorithms, i.e., SVM, artificial neural network (ANN), linear discriminant analysis (LDA), and K-nearest neighbors (KNN), to predict the EMG features to five types of hand movements, and then used the classification result as an input of the assistance controller. The results showed that SVM had the best accuracy among those four in predicting the type of hand movements. To improve the performance of the classification model, studies such as [8,50] compared two types of classifiers with same algorithm but different input signals. The type 1 classifier used a single channel EMG signal but converted it to 14 different statistical features; the type 2 classifier used EMG signal from five channels and only converted them to a single statistical feature. Both type 1 and 2 classifiers were designed to detect if the arm was in motion or not. The results indicated that the type 1 classifier had higher accuracy, but the type 2 classifier had less latency. Refs. [8,50] indicated that extracting the EMG signal to multiple features can improve the accuracy of a classification model. Another two studies [28,52] used a sensor fusion method that used both EMG and electroencephalography (EEG) signals to improve the accuracy of the classification model, and [46] implemented a threshold method in which the signal amplitudes must be greater than a predefined threshold in order to be input to the classification model. Such a threshold method can prevent misclassification and can improve the accuracy of myoelectric control systems.

### 2. Regression-based Myoelectric Control

The regression model represents the relationship between input and output data as a function that is trained with a pre-collected dataset. Compared to the classification model, which can only detect discrete motions, the regression model can output continuous variables with the statistical feature of the EMG signal as input, such as joint angle and joint torque. The regression models can be constructed using different approaches, such as linear regression, ANN [27], and Kalman Filters [57]. Because of the nonlinearity of the EMG signal, the articles included in this review only used ANN or the Kalman Filter as the regression model.

Among the included articles, ref. [27] implemented a machine learning-based myoelectric control system to control an elbow exoskeleton by training a back propagation neural network (BPNN) to estimate joint angle from the statistical feature of the user's EMG signal, and showed that the regression model could accurately estimate the user's joint angle. Another application of such machine learning-based myoelectric control systems was bilateral hand training with a wearable hand exoskeleton, which used the statistical feature of an EMG signal from an unimpaired hand to estimate the magnitude of the assistive force, to train the impaired hand. In this review, research articles such as [18,20,22,32,46,56,69], used neural network-based regression models as a myoelectric control system to control a hand exoskeleton for training different parts of the impaired hand. On the other hand, ref. [57] used a Kalman filter-based regression model to compute the joint torque based on the EMG signal. Compared to the neural network model, the Kalman filter model does not need much time, nor extensive datasets to train the model. Moreover, tuning the Kalman filter model for different users only requires the measurements of a few sets of joint torque under different positions. Although the regression model in [57] reduced the effort in training the model, its accuracy and efficacy in assisting the upper limb were similar to that of the neural network-based regression model. Based on the adaptive and robust regression model developed in [57], ref. [52] integrated a neural network and a Kalman filter-based regression model in which the neural network took multiple variables as inputs, which includes processed EMG signals, the joint torque estimated by the Kalman filter, and the joint angle and joint angular velocity measured by the IMU sensor, to control the motion of the upper limb exoskeleton. The myoelectric control system proposed in [52] achieved a better accuracy and efficacy compared to the previous regression models explained above.

3. Reinforcement Learning-based Myoelectric Control

In addition to the classification and regression model, the reinforcement learning model was also used in the myoelectric control system in the reviewed articles. Unlike the classification and regression models, the reinforcement learning model is trained with a reinforcement learning algorithm that uses a smart agent to learn the optimal policy while interacting with an environment. During the process of reinforcement learning, the smart agent exerts an action on the environment, based on its observation of the environment (state) as feedback, and then the environment returns a score to evaluate the agent's action (reward). Based on the reward returned from the environment, the agent optimizes its policy toward the direction of greater reward [85]. In the field of wearable robotics, reinforcement learning has been used in prosthetic control (e.g., [86]), lower limb exoskeleton control (e.g., [87]), and joint torque estimation (e.g., [88]). Compared to other types of control methods used in EMGs, the reinforcement learning algorithm reflects the interaction between humans and the environment. In addition, reinforcement learning can provide an optimal control policy without the knowledge of the environment, which is ideal for use in complex and uncertain environments. Hamaya et al. [40] used a reinforcement learning algorithm called Probabilistic Inference for Learning Control (PILCO) to control an elbow exoskeleton. The state vector consisted of the kinematics of the elbow joint and EMG signals, and the reward was related to the difference between the desired and actual trajectory. The PILCO algorithm implemented the Gaussian process to learn the probabilistic dynamic model of the human-exoskeleton interface through the states collected during human-exoskeleton interaction, then evaluated the control policy using the learned probabilistic dynamic model, and finally optimized the control policy through the policy gradient method [89]. This method offered a faster training time as compared to other machine learning myoelectric control systems. As mentioned above, the ML-based myoelectric control schemes show promising results in increasing the accuracy and reliability of myoelectric control of an upper limb wearable robotic exoskeleton. However, some limitations exist in the ML-based myoelectric controls, such as a limited implementation in multi-DOF upper limb wearable robotic exoskeletons [9], poor performance in online training of the machine learning model [90], and high computational cost [53].

#### 3.2.5. Neural-Fuzzy Myoelectric Control

The neural-fuzzy-based myoelectric control system is another scheme for controlling an upper limb exoskeleton-type exoskeleton. According to the theory of modern artificial intelligence, neural fuzzy is defined as the combination of a neural network and fuzzy logic [91]. The fuzzy logic outputs the control command while the neural network tunes the fuzzy logic iteratively, which allows for training of the fuzzy logic during use (i.e., fuzzy modifier) (as shown in Figure 9).



**Figure 9.** The conceptual block diagram for neural-fuzzy-/fuzzy-logic-based myoelectric control system.

José et al. [92,93] first introduced the neuro-fuzzy myoelectric control system, where the fuzzy logic consisted of "IF premise set> and THEN <consequent set>" statements,

and the fuzzy modifier was a fully connected neural network. The premise set in this fuzzy logic consisted of the motion status of EMG signals from eight muscles, and the consequent set included the desired motor torque and the gain of controllers. The fuzzy logic needed to be tuned by the neural network using the EMG signals. The tuning process was similar to training a neural network that uses backpropagation and activation functions. Since the fuzzy logic is trained each time before making a control command, it makes it adaptive to various environments. This type of neuro-fuzzy myoelectric control system has been adopted by [16,92], and the results have shown that it reduces the user's muscle activation during various activities of daily life. However, the notable downside of this approach is that the fuzzy logic becomes complicated as the number of DOFs of the exoskeleton increases. When a exoskeleton has multiple DOFs, the number of fuzzy logics will increase exponentially, consequently increasing the difficulty in implementing and training fuzzy logics.

To improve the compatibility of the neuro-fuzzy myoelectric control system for higher DOF exoskeletons, a different type of fuzzy logic was implemented on a 7-DOF upper-limb exoskeleton [17], which represented fuzzy logic as a matrix product of the EMG signal from 16 channels and the weights of each neuron in the fuzzy modifier. Similar to other studies, the fuzzy modifier was a fully connected neural network tuned every time before running. The experiment of this study showed that under the assistance of a wearable robotic exoskeleton, the muscular efforts of each muscle segment were reduced.

#### 3.3. Key Design Characteristics

This section presents the literature research for the design aspects of upper limb exoskeletons that are equipped with the myoelectric control system, which include degrees of freedom, portability, and the application of upper limb exoskeletons.

#### 3.3.1. Degrees of Freedom

As shown in Figure 10a, 13% (seven articles) of the studied articles have worked on controlling two DoF systems. Additionally, 3%, 15%, and 27% (1, 9, and 16 articles, respectively) of the explored studies pertain to the three, four, and greater than four DoF robots. Considering the single DoF exoskeletons (42% of the reviewed papers and 25 articles), many have exclusively studied the flexion and extension of the elbow joint (Table 2). During testing protocols, they may require the subjects to make different selected angles wearing the arm device while handling varying load weights [9,35]. The biceps muscle responsible for elbow flexion and extension motion would then be used as the control signal. In some studies, while flexion actuation is achieved through the device, gravity is responsible for extension actuation [23,25]. Many papers have focused on the actions of the fingers or that of the wrist (Table 2), within their robotic hand device. A hand exoskeleton can be activated simply through two independent DoFs, one for the thumb and one for the four fingers' flexion/extension [65]. Alternatively, for grasping movement, the thumb can passively be fixed to its position [20,32]. Soft robotic gloves are also a popular field of research, where the production of sufficient forces to grasp objects through the robotic glove is studied [19].

#### 3.3.2. Portability of Upper Limb Robotic Wearable Exoskeletons

Among the reviewed literature, 63% (37 articles) of studies developed upper limb exoskeletons with portable structures, while the remaining ones presented fixed upper limb robots (as shown in Figure 10c). Portability is an important design requirement for exoskeletons that have intended applications in the performance augmentation of able-bodied individuals or for the assistance of older adults and patients with neuromusculoskeletal disorders in activities of daily life. For rehabilitative use, exoskeletons need not be essentially portable if they are being utilized in clinical environments or therapeutic institutes. However, the portability of rehabilitative robots is desired for home-based rehabilitation en-



vironments since they can provide ease of access and more flexibility to the user, compared to stationary robotic systems.

**Figure 10.** (a) The percentage of the degrees of freedom for the upper limb exoskeletons in the included research literature; (b) The percentage of the applications of the upper limb exoskeletons in the included research literature; (c) The percentage of portability of the upper limb exoskeletons in the included research literature.

The majority of studies that presented portable upper limb robots were based on hand exoskeletons [15,18–20,22,32,41–44,46,48,49,54–56,65,69,72]. There were also a number of studies that developed portable exoskeletons/exosuits to support elbow movement [17,23,27,31,38,39,45,52,57–61,64], or both hand and elbow motions [24,28,51]. Only one portable exoskeleton was found in the literature that provided power assistance for the flexion/extension movement of both elbow and shoulder joints [71]. Most of the portable upper limb robots mentioned above had self-contained architecture. Only a few powered exoskeletons assisting elbow movement [24] or elbow–shoulder movements [71] required the user to wear backpack support while operating the device.

When an upper limb exoskeleton is designed to support movements of more than one joint, the number of required actuators and the weight of the robot both increase. Therefore, most of the exoskeletons supporting motions of two or multiple joints, such as shoulder–elbow [29,47], elbow–wrist [8,26], or shoulder–elbow–wrist [16,17], found in the literature, were not portable. Apart from that, some studies also mentioned fixed exoskeletons supporting a single joint motion, which were mainly elbow exoskeletons [9,17,21,25,30,33,34,37,40,47,53,62,63] (except one shoulder exoskeleton in [68] and one hand exoskeleton in [36]), which were not portable because of their high weight-to-power ratio.

#### 3.3.3. Application of Upper Limb Robotic Wearable Exoskeletons

The potential application scenarios identified in the reviewed literature can be broadly categorized as rehabilitative use (58%, 34 articles), assistive use (30%, 18 articles), and human augmentation (12%, eight articles) (as shown in Figure 10b). Rehabilitative upper limb exoskeletons can aid patients with motor disabilities in performing therapeutic exercises, in order to restore/improve the motor functions in their impaired limbs. A total of 58% (34 articles) of the exoskeletons in the reviewed papers were intended for rehabilitation scenarios, assisting the patient in performing different single or multiple joint motions

(some examples include [15,16,21,24,29,68]). The benefits of wearable robot-assisted rehabilitation compared to traditional physical therapy include the provision of repetitive and intensive training to the patient, lessening the physical burden of the therapist, and offering an effective means of objectively quantifying the patient's progress before and after training [1].

About 30% (18 articles) of the reviewed literature developed assistive upper limb wearable robotic exoskeletons (e.g., [22,26,36,41]), which were intended to support people with neuro-muscular conditions (such as stroke, spinal cord injury, muscle weakness, etc.) as well as older adults in performing the activities of daily living (ADLs). Such assistive robots may enable users to regain some of their functional independence, facilitate their participation in daily activities and overall, enhance the quality of their lives. Furthermore, some wearable robotic exoskeletons (e.g., [17,35,63]) have been designed for rehabilitation and assistance scenarios to be utilized in physical therapy in clinical environments, or to provide assistance during ADLs.

Only 12% (eight articles) of the reviewed exoskeletons were developed for potential application in human performance augmentation [31,39,50,71], which can enhance the strength, endurance, or physical capabilities of healthy individuals during repetitive and/or strenuous tasks in factories, warehouses, or military bases or excursions. Using these devices can reduce the risk of developing musculoskeletal disorders in industry workers and military personnel, as well as reducing their metabolic costs during lifting or carrying of heavy loads.

#### 3.4. Human-Subject Evaluation of the Myoelectric Control System

Among the reviewed literature, nine studies tested their prototypes on subjects with neuro-muscular impairment (either stroke or SCI patients) [15,24,32,36,41,54,64,65], while the remaining ones performed experimental validations on healthy subjects. The validation tests found in the reviewed studies can be broadly grouped into the following categories: (i) kinematic evaluation, (ii) user's effort evaluation, (iii) ML model performance evaluation, and (iv) clinical assessment.

The controller performance was tested in many studies using kinematic evaluation, where the most commonly used outcome measures identified were trajectory tracking performance (i.e., joint angle tracking) and velocity [9,16,17,21,22,24,57,61,62,65,67]. Several studies tested the proposed controller's efficacy by evaluating the user's effort in performing a specific movement or task [9,17,23,57,60,63]. To do so, EMG signals from relevant muscles were measured during static/dynamic tasks to estimate muscle activation and EMG-derived joint torque. Then, comparisons were made among different scenarios (such as not wearing the exoskeleton, wearing it without assistance, and wearing it with assistance under the same or different loads) to quantify the reduction in muscle effort. In addition, a few studies utilized isometric contraction tests to evaluate the effect of the developed prototype on muscle fatigue [23,64]. Classification accuracy and confusion matrix outcomes commonly evaluated the ML-based myoelectric control systems' performances and computation times (or latencies) [18,63,64,67–69]. Only one study evaluated the learning time for a model-based reinforcement learning framework, and found that the proposed method could learn the proper assistive strategy after only 60 seconds of user-robot interaction [40].

Only one study [15] performed a clinical assessment on eight chronic stroke subjects using the Fugl-Meyer Assessment (FMA) and the Action Research Arm Test (ARAT) to evaluate the improvement in hand and upper arm functions, respectively, after 20 sessions of hand exoskeleton-assisted training. In addition, the Sollerman Hand Function Test was used in another study [41] to evaluate the hand performance of two SCI patients with and without wearing the active hand exoskeleton.

### 4. Discussion

In this review, myoelectric control systems designed, implemented, and validated on upper limb exoskeletons, especially the rigid-link exoskeleton and exosuit, were classified into threshold-based, proportional, biomechanical model-based, machine learning-based, and neural-fuzzy myoelectric control systems. In addition, each control modality provided details on the control methods, performance, and limitations. These existing myoelectric control systems show promising outcomes, demonstrating enhanced human-robot interactions, robot intelligence, and adaptiveness to the user, task, and environment compared to traditional control systems without using electromyography. However, several challenges and limitations are commonly discussed and explored in the articles associated with myoelectric control systems applied to upper limb wearable robots. The limitations and challenges extracted from the included research articles suggest that future studies of myoelectric control systems for upper limb exoskeletons should focus on narrowing the gap between laboratory studies and clinical applications. Further studies are needed to investigate the myoelectric control system of upper limb exoskeletons in clinical environment. This chapter discusses the research questions and tasks for future research works to narrow the gap between laboratory studies and clinical applications, which include (1) the robustness of machine learning-based myoelectric control, (2) the calibration procedure of the biomechanical model-based myoelectric control system, (3) the implementation of myoelectric control systems to a high degree-of-freedom wearable exoskeleton, (4) the incorporation of safety requirements in myoelectric control systems, (5) a clinical assessment of assistive and rehabilitative wearable exoskeletons based on myoelectric control systems, and (6) the incorporation of human-centered myoelectric control. These are crucial barriers to the effective use of myoelectric controls on upper limb wearable exoskeletons, warranting further research.

## 4.1. *How Can We Improve the Robustness of Machine Learning-Based Myoelectric Control Systems?*

The robustness of the myoelectric control system relates to its capability to resist disturbance from and within the environment on electromyography signal [94]; such a disturbance is generally caused by muscle fatigue [95], electrode shift [96], and changes in EMG patterns over time [97]. The number of studies using machine learning (ML)-based myoelectric control systems has significantly increased over the past decade, showing promising performance from preliminary/pilot testing in laboratory settings. However, none of these ML-based myoelectric control systems have yet studied approaches to improving their robustness. To bridge the gap between experimental research and commercial/clinical applications, machine-learning-based myoelectric control systems should focus on developing an accurate control scheme under a well-controlled laboratory environment and improving the robustness under real-world environment [98].

Within the reviewed research articles that implemented machine-learning-based myoelectric control systems in this review, commonly reported issues include varying characteristics of EMG signals between different physiological conditions, noise/artifacts, and muscle fatigue that cause variants in the EMG signal, and electrode shifts during or between sessions. However, none of the research articles included in this review specifically addressed these issues. Existing studies on the myoelectric control of prosthetics, teleoperate robotic arms, and the pattern recognition of EMG signals have already investigated such issues. For example, [99] explored the property of muscle coordination in the frequency domain, and introduced a novel feature of the EMG signal to improve the accuracy in classifying movement intention from the EMG signal to incorporate variations in the EMG signal. The authors in [100] used a post-processing method to enhance the robustness of the classifier against the variant in the EMG signal caused by muscle fatigue and noise/artifacts. The authors in [101] investigated the optimal distance between EMG electrodes and the feature selection method for myoelectric pattern recognition under the impact of electrode shift, and provided solutions to improving the robustness of myoelectric pattern recognition in the presence of an electrode shift [94]. The behaviors of all time and frequency domain EMG features are evaluated to classify upper limb motions. Based on the abovementioned research, the potential approaches for improving the robustness of machine learning-based myoelectric control systems include the use of more efficient features, reducing the impact of EMG electrode shifts, and improving the data collection protocol or signal processing method to enhance the robustness of the machine learning model. Further investigation of the combination of the abovementioned robustness-improving methods and myoelectric control systems and their performance on the upper limb exoskeleton are needed. One of the open questions is to investigate the performance of an upper limb exoskeleton with a machine learning-based myoelectric control system while using different time-domain and frequency-domain features. This is because the results of our literature research indicate that most of the included research articles use the common types of features such as root mean square (RMS), mean absolute value (MAV), and zero crossing (ZC). Instead, the selection of EMG features for future studies should be expanded to all time-domain and frequency-domain features [102] and evaluate the performance of the human-exoskeleton system. With the improved myoelectric control system, the effectiveness of assisting the patient in regaining upper limb functionalities in rehabilitative training will be increased, as well as the potentiality for reducing the occurrence of musculoskeletal disorders. Furthermore, a possible shifting of EMG electrodes during use and its associated effects on myoelectric control performance should be examined. For example, while training and evaluating machine learning-based myoelectric control, the EMG electrodes can be placed slightly differently to improve the overall robustness of the model. Moreover, training protocols for a machine learning-based myoelectric control system is also less well studied For example, [103] used EMG signals collected in 21 days to train a classification model that showed that the EMG signal collected in the long term even decreased the accuracy of classification. This suggests that there is a need to investigate training protocols that generate the most optimal dataset to train machine learning-based myoelectric control systems.

In addition to the above mentioned research fields, future research should also explore the implementation of sensor fusion methods and their contributions to improving the robustness of machine learning-based myoelectric control systems. The sensor fusion of myoelectric control systems refers to the combination of sensors different from the EMG sensors to achieve a better accuracy and a better inference on the user's movement intention. In this review, some research articles implemented sensor fusion methods to improve the accuracy of the myoelectric control system [28,30,45,52]. Some articles have benchmarked the performances of myoelectric control systems with sensor fusion by comparing them with those without sensor fusion [28]. Among the included articles that have studied sensor fusion methods, the human-robot interactive force measured by the FSR sensor, joint kinematics measured by the IMU sensor, and brain signal measured by electroencephalogram (EEG) sensor were the major signals that were combined with the EMG sensor. Identifying the optimal set of different sensor inputs and utilizing sensor fusion will lead to an enhancement of the accuracy and robustness of machine learningbased myoelectric control system. Furthermore, all such studies implemented the sensor fusion method for the machine-learning-based myoelectric control system, and the results have demonstrated the superior performance of myoelectric control systems with the sensor fusion method. However, not only the machine-learning-based myoelectric control system, but other types of myoelectric control can also work with sensor fusion methods control systems [63,104].

## 4.2. *How Can We Improve the Calibration Procedure of Biomechanical Model-Based Myoelectric Control Systems?*

The parameters of the human musculoskeletal models implemented in myoelectric control systems vary between users. Therefore, repetitive customization and calibration are required before each use, significantly affecting the practicality of wearable exoskele-

tons using biomechanical model-based control, and limiting their translation from in-lab assessment to real-world applications. The calibration of the musculoskeletal model needs additional experiments before its use as the anatomical model of the limb. For example, in [59], a motion capture system was utilized to facilitate the calibration procedure. First, the kinematic data of users were collected using the motion capture system while performing the designated motions. Then, a biomechanical simulator (OpenSim) was used to scale a generic musculoskeletal model with the captured data to calibrate the biomechanical model in the myoelectric control system. This process consumes a significant amount of time and cannot be conducted by the users themselves. Some included articles investigated different approaches to simplify the calibration procedures; for example, ref. [29] implemented a data-driven optimization method to calibrate the musculoskeletal model, and [47] measured the gravity and interaction torque under various load conditions and used them to optimize the parameters of the musculoskeletal model using the Genetic Algorithm. Both approaches have demonstrated much easier calibration procedures than model-based myoelectric control systems within the included articles [23–25,33,35,37,59,63,72].

Although the included articles have made many efforts to simplify the calibration procedure [29,47] and significant improvements have been achieved, implementing the generic musculoskeletal model to the myoelectric control system of the upper limb exoskeleton has not been explored. Unlike the subject-specific musculoskeletal model, which is calibrated for a single human user, the generic musculoskeletal model is calibrated using the data collected from a group of humans. Compared to the subject-specific musculoskeletal model, the generic musculoskeletal model is less accurate but has better adaptability to different human users. However, Turvey [105] studies mentioned that the performance of a generic musculoskeletal model-based myoelectric control system for a simulated upper limb prothesis was similar to the myoelectric control system with a subject-specific musculoskeletal model. The factors contributing to the reliability of generic musculoskeletal model-based myoelectric control are still unclear. However, possible factors include (1) co-adaption between human and myoelectric control systems and (2) conserved biomechanical action during motions. Therefore, instead of focusing on how to improve the calibration procedure of biomechanical model-based myoelectric control systems, future studies could also investigate the questions relevant to the generic musculoskeletal model for biomechanical model-based myoelectric control systems of upper limb exoskeletons; for example, the researchers can use different approaches to form a group of human subjects to collect the biomechanical data for a generic musculoskeletal model, and to investigate the impact of various human subject groups on the accuracy of the generic musculoskeletal model. Furthermore, future studies could also study the co-adaption between human users and biomechanical model-based myoelectric control systems. Only [9] within the selected articles studies the co-adaption between the human user and the proportional myoelectric control system, and the co-adaption between the human user and biomechanical modelbased myoelectric control system is still unexplored. However, as the exoskeleton is a symbiotic robot that requires a close interaction between humans and the exoskeleton, understanding the co-adaption between human and biomechanical model-based myoelectric control systems can help researchers to improve the robustness and adaptability of myoelectric control systems, and also to accelerate the development of generic musculoskeletal model-based myoelectric control systems of upper limb exoskeletons.

## 4.3. *How Can We Implement Myoelectric Control Systems for High Degree-of-Freedom (DOF) Upper Limb Exoskeletons?*

Most papers have explored only a single DoF control scheme, rather than the simultaneous control of multiple DoFs, due to its relative simplicity, while a multiple DoF system could render the enhanced adaptability of the robot. Compared to a single DoF actuation control, the simultaneous control of multiple DoF is challenging in practice. Having more than a single DoF requires the exoskeleton joint motion controller to properly coordinate the desired limb motion. During complex motions, including cyclic tasks (e.g., table wiping or cutting vegetables), the frequencies of different joint movement trajectories must be synchronized to produce the robot's desired periodic motion. However, in the conventional EMG-based control methods, e.g., where a single (or a pair of) muscle activity is mapped to a single DoF actuation, joints with only a few DoF can be controlled because of their low motion complexity. Similarly, a fuzzy controller's rules would become convoluted with an increased number of DoF and/or periodic motions. For the more involved and higher DoF movements, such as that of the hand, a machine learning-based myoelectric muscle motion pattern recognition has proven to be a reliable method due to its characteristics previously detailed in Section 3.3.2 [54]. In addition, an EMG signal is usually not limited to the data generated by one muscle, and could also be reflecting the activity of the neighboring muscles in the area, the so-called crosstalk issue [106]. Therefore, the fact that different muscles are associated with different motions leads to difficulty in the unequivocal association of the EMG signal with motion. Estimating a single joint angle with little influence of a change in other joints' positions and reducing crosstalk is the best practice [24]. Additionally, high computational costs remain another barrier in the physical implementation of such detection methods into real-time embedded systems.

To provide effective assistance in ADLs, an upper limb wearable exoskeleton must have a high number of DoFs, and a controller to coordinate those motions while requiring the user to put in a minimum of effort. Therefore, dimensionally reducing the control problem can be a possible approach. In one study, to dimensionally reduce the essential control inputs, a 10-DoF (independently actuated) whole-hand exoskeleton took advantage of kinematic synergies—a collection of relatively independent degrees of freedom behaving as one functional unit [105]—enabling the researchers to test different synergies or control approaches [56]. They experimented with three synergies related to the thumb, index, and remaining three fingers' independent flexions/extensions. Researchers have successfully controlled the hand exoskeletons to perform common hand gestures for all the fingers, excluding the thumb, as the complex motion of the thumb imposes difficulties on the mechanical structure design [44,65]. It remains a challenge to design mechanisms for all fingers to build a complete hand exoskeleton that provides rehabilitation for different grasping movements.

Finally, the number of actuators increases with the number of DoFs, causing limitations in wearability and portability as it increases the weight, cost, and size. Therefore, reducing the number of DoFs; for example, mechanically coupling a few joints with small ranges of motions or requiring low actuation forces, can make the device simpler and more lightweight. One drawback of this solution is that it can limit the flexibility and possible motion patterns of the system. Another possible solution can be having fewer actuators than the number of DoFs, which is known as the under actuation strategy [44].

#### 4.4. How to Incorporate Safety Requirements in Myoelectric Control Systems?

Safety is a critical requirement for upper limb exoskeletons, particularly for active powered systems. For assistive and rehabilitative applications, it is essential to ensure the safety of the patients so that these robots will not cause a risk of injury during use. Existing exoskeletons in the literature mainly focus on implementing safety measures in mechanical design, such as installing mechanical stops, rotation limits, or force limits to guarantee that there is no excessive range of motion movements or excessive force applied to the user [2].

However, such mechanisms cannot always fully warrant the safety of the user in the presence of unknown large parameter variances, hardware failures, or actuator defects [107]. Therefore, control strategies that can effectively compensate for different uncertainties and external load disturbances might substantially contribute towards improving the safety of the user while performing motions/tasks wearing the robotic exoskeleton. Given the inherent variability of EMG signals (arising from changes in arm posture, electrode repositioning, fatigue etc.), one possible approach can involve applying data fusion techniques on EMG signals to reduce the potential errors of motion estimation. In [108], two data fusing algorithms, Variance Weighted Average (VWA) and Decentralized Kalman Filter

(DKF), were proposed for the myoelectric control of a robotic arm, in order to improve its reliability under electrode faults and noisy environments.

For machine learning- or deep learning-based myoelectric control systems, implementing post-processing techniques, such as multiwindow smoothing and confidence estimation, may also improve the reliability of the controller [109]. Multiwindow smoothing approaches can reduce potential estimation errors by utilizing EMG signals from successive sliding windows [110], while confidence estimation is used to analyze the confidence of classification results so that uncertain decisions can be detected and eliminated [111]. Furthermore, using adaptive control with real-time learning and prediction can also help in increasing the fault tolerance of the proposed system [112]. Future studies may consider focusing on how to improve the safety measures in wearable exoskeletons from both mechanical and controller design perspectives to enhance the reliability and fault-tolerance of upper limb exoskeletons for real-world applications.

## 4.5. How Can We Improve Experimental Validation and Clinical Assessment Methods for Assistive and Rehabilitative Exoskeletons Based on Myoelectric Control Systems?

The myoelectric control systems reviewed in the literature were tested in laboratory settings, mostly on able-bodied users. Although many studies showed good preliminary results with healthy subjects, to transition from laboratory to real-world assistive and rehabilitative applications, these prototypes need to be further tested on large-scale clinical populations to address inter- and intra-subject variability issues. Because the voluntary muscle contraction level and muscle fatigue endurance may vary between healthy and neurologically impaired subjects; thus, it might impact on the efficacy of the proposed control method in assisting the intended motion/task of subjects with motor disabilities with high accuracy and reliability, despite achieving desired results with healthy users. For example, in [36], the overall average controller accuracy for a hand exoskeleton was  $98.1 \pm 4.9\%$  among neurologically intact subjects. However, when tested on SCI subjects, the controller accuracy was found to be lower (90.0  $\pm$  13.9%). Similarly, in [64], the average classification accuracy of the proposed EMG-based classifier for an upper limb exoskeleton was lower among the SCI group (85%-95% for single-DoF and 60% for multi-DoF) compared to able-bodied participants (99% for single-DoF and 90% for multi-DoF). These findings highlight the importance of experimental validation of myoelectric control systems on a clinical population to evaluate the controller performance in the presence of neurological impairment. Furthermore, while interpreting study outcomes, the different level of motor impairment among clinical subjects should also be taken into account (as performed in [32]), particularly in those cases where patients with different neurological conditions or motor function levels are recruited for testing.

Experimental validation methods for rehabilitative exoskeletons in the reviewed literature have mostly been limited to evaluating trajectory tracking performance, movement execution velocity, user's effort, and joint torque estimation for one session of lab assessment. Although these outcome measures may indicate the functionality of the physical model and the efficacy of the control system in assisting the user in performing a certain movement or task, it might not completely reflect the effectiveness of the robot-assisted training in improving the motor function of subjects with neuro-muscular disorders. Since clinical validation studies of myoelectric control systems for upper limb exoskeletons are still in their infancy, the focus should be given to performing experimental validation with control groups at least, to provide stronger evidence for the effectiveness of robot-assisted training. In addition, there is a possibility that the classification accuracy of ML-based myoelectric control might not be strongly associated with the training effectiveness. For example, in [113], the real-time testing of myoelectric control prosthetics on trans-radial amputees with virtual arm showed that the completion rate of predefined hand motions was 55% on average, despite obtaining 85% classification accuracy for the pattern recognition method. This suggests that this proposed pattern recognition-based myoelectric control was not able to reliably control real-time hand grasp, even with good classification accuracy. Clinical assessment tests are useful in evaluating the efficacy of myoelectric controlbased exoskeletons in achieving desired outcomes among neurologically impaired patients. For example, in [41], the Sollerman hand function test was used to assess the performance of a hand exoskeleton in improving hand functions required for common ADLs among SCI subjects. Also, in [15], the Fugl-Meyer Assessment (FMA) and the Action Research Arm Test (ARAT) were used to evaluate the effect of an upper limb exoskeleton on upper limb functions and hand functions among chronic stroke patients, respectively. The results showed that the motor functions of the subjects on the hand and upper limb significantly improved after 20 sessions of robot-assisted hand functions task training. Therefore, future studies should consider incorporating multiple robot-assisted training sessions, followed by clinical assessment tests, to evaluate the robustness and effectiveness of rehabilitative upper limb exoskeletons in restoring upper limb functions in patients with motor disabilities.

## 4.6. How Can We Develop a Human-Centered Myoelectric Control System Instead of a Task-Centered Myoelectric Control System?

The research literature included in this systematic review has shown innovative ideas for using EMG signals to control an upper limb wearable robot. However, the presented control modalities can be broadly categorized as using a machine learning model or a musculoskeletal model, using EMG for the continuous control of assistive torque (proportional), and starting the assistance of a wearable robot if the EMG signal satisfies certain conditions (neural fuzzy and threshold). Despite these articles demonstrating promising results, these myoelectric control systems are developed to map the user's movement intention from the EMG signal to a specific task, and then to control the wearable robot to complete this task. The wearable robots controlled by such types of myoelectric control systems only focus on finishing the assigned tasks, and are unable to perceive the feedback of the user. However, the wearable robot and human physically interact in a bidirectional approach; therefore, understanding the feedback of the human by developing a human-based myoelectric control system can improve adaption and synchrony between the human and the wearable robot.

Different from the existing literature, the human-based myoelectric control system should focus on fulfilling the human's motion intention, which can be achieved by minimizing human effort (e.g., metabolic cost, muscle activation/fatigue, etc.). For example, when controlling an upper limb wearable robot to help the human to lift a heavy load, the myoelectric control system should no longer estimate the load from the EMG signal and control the robot to output the corresponding assistive force to compensate for the gravitational load. Instead, the myoelectric control system will control the wearable robot to output a proper amount of assistive force to minimize the amplitude of the human's EMG signal. Compared to the conventional control approach, the human-based myoelectric control system is adaptive to the variant of EMG signal, which reduces the dependency on calibration. As a matter of fact, such an approach has been introduced by [114] to control an upper limb wearable robot to compensate for the gravity of the human arm. However, ref. [114] only validates the proposed control system in a simulation environment. Within the included literatures, ref. [40] used a model-based reinforcement learning algorithm to learn the control strategy of the upper limb wearable by minimizing the amplitude of EMG signal. However, such a model-based reinforcement learning algorithm requires a significant amount of computational power. Furthermore, ref. [35] studied a myoelectric control system based on a similar concept, but used the musculoskeletal model to convert the EMG signal to the muscle force before using it as a control command. Moreover, the abovementioned myoelectric control systems are only developed for human power augmentation by compensating gravitational loads. Therefore, the applications of human-based myoelectric control are largely unexplored. Further studies should consider exploring the application of such control schemes on more dexterous motions, more complicated scenarios such as rehabilitative training, and the control system that requires less computational power.

## 5. Conclusions

The myoelectric control systems have shown their efficacy for use in the upper limb exoskeletons to enhance human-robot dynamics and interactions. This review focused on identifying and categorizing key control strategies employed in myoelectric control systems implemented on upper limb exoskeletons. The overview of each myoelectric control system was presented, highlighting their control performance and accuracy, as well as their limitations and challenges. Some future research directions are proposed: improving adaptiveness across different users, tasks, and environments, myoelectric control systems for multi-joints and/or bilateral arm assistance, improving robustness against the disturbance in the real-world application, developing a human-centered myoelectric control system, and improving its safety features through a human subject evaluation process. It was noted that comparably less work has been performed to exploit a human-centered myoelectric control system, and fewer validated models exist in the literature. This warrants further research to assess the applicability of a human-centered myoelectric control system to control upper limb wearable exoskeletons. Further, all myoelectric control systems discussed in this review were evaluated in laboratory settings, and mostly on healthy able-bodied individuals, limiting their implications on clinical applications. Nevertheless, the rapid evolution and the many variations of myoelectric control systems observed in recent years suggest that it is the most promising strategy for the intelligent, adaptive, robust, and human-in-loop control of upper limb exoskeletons.

Author Contributions: Conceptualization and methodology: All Authors; investigation: J.F.; data collection and validation: J.F., R.C. and S.M.H.; writing—Introduction and Methods: J.F.; writing—Results overview: J.F. and R.C.; writing—Results control method: J.F.; writing—Results DoF: S.M.H.; writing—Results portability: R.S. and R.C.; writing—Results application: R.C.; writing—Results evaluation: R.C.; writing—Discussion: All Authors; writing—Conclusion: J.-H.P.; writing—proofreading and edition: J.F., R.C. and J.-H.P.; corresponding author—J.-H.P. All authors have read and agree to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

**Informed Consent Statement:** Not applicable.

Data Availability Statement: Not applicable.

**Conflicts of Interest:** The authors declare no conflict of interest.

### References

- 1. Bardi, E.; Gandolla, M.; Braghin, F.; Resta, F.; Pedrocchi, A.L.; Ambrosini, E. Upper limb soft robotic wearable devices: A systematic review. *J. NeuroEng. Rehabil.* **2022**, *19*, 1–17. [CrossRef] [PubMed]
- 2. Gull, M.A.; Bai, S.; Bak, T. A Review on Design of Upper Limb Exoskeletons. Robotics 2020, 9, 16. [CrossRef]
- Robertson, D.G.E.; Caldwell, G.E.; Hamill, J.; Kamen, G.; Whittlesey, S.N. Research Methods in Biomechanics; Human Kinetics: Champaign, IL, USA, 2014.
- 4. Abas, N.; Bukhari, W.M.; Abas, M.A.; Tokhi, M.O.; IEEE. Electromyography Assessment of Forearm Muscles: Towards the Control of Exoskeleton Hand. In Proceedings of the 2018 5th International Conference on Control, Decision and Information Technologies (CODIT), Thessaloniki, Greece, 10–13 April 2018; pp. 822–828. [CrossRef]
- 5. Rechy-Ramirez, E.J.; Hu, H. Bio-signal based control in assistive robots: A survey. *Digit. Commun. Netw.* **2015**, *1*, 85–101. [CrossRef]
- 6. Bi, L.; Feleke, A.G.; Guan, C. A review on EMG-based motor intention prediction of continuous human upper limb motion for human-robot collaboration. *Biomed. Signal Process. Control* **2019**, *51*, 113–127. [CrossRef]
- Lotti, N.; Xiloyannis, M.; Missiroli, F.; Chiaradia, D.; Frisoli, A.; Sanguineti, V.; Masia, L. Intention-detection strategies for upper limb exosuits: Model-based myoelectric vs dynamic-based control. In Proceedings of the 2020 8th IEEE RAS/EMBS International Conference for Biomedical Robotics and Biomechatronics (BioRob), New York, NY, USA, 15 October 2020. [CrossRef]
- Accogli, A.; Grazi, L.; Crea, S.; Panarese, A.; Carpaneto, J.; Vitiello, N.; Micera, S. EMG-Based Detection of User's Intentions for Human-Machine Shared Control of an Assistive Upper-Limb Exoskeleton. In Wearable Robotics. Proceedings of the 2nd International Symposium on Wearable Robotics, WeRob2016, Segovia, Spain, 18–21 October 2016, Challenges and Trends / José Gonzalez-Vargas [and Four Others]; González, J., Ed.; Springer: Cham, Switzerland, 2017; Volume 16.

- Lenzi, T.; Rossi, S.M.M.; Vitiello, N.; Carrozza, M.C. Intention-based EMG control for powered exoskeletons. *IEEE Trans. Bio-Med Eng.* 2012, 59, 2180–2190. [CrossRef]
- 10. Roche, A.D.; Rehbaum, H.; Farina, D.; Aszmann, O.C. Prosthetic myoelectric control strategies: A clinical perspective. *Curr. Surg. Rep.* **2014**, *2*, 1–11. [CrossRef]
- 11. Vélez-Guerrero, M.A.; Callejas-Cuervo, M.; Mazzoleni, S. Artificial intelligence-based wearable robotic exoskeletons for upper limb rehabilitation: A review. *Sensors* **2021**, *21*, 2146. [CrossRef]
- Liberati, A.; Altman, D.G.; Tetzlaff, J.; Mulrow, C.; Gøtzsche, P.C.; Ioannidis, J.P.A.; Clarke, M.; Devereaux, P.J.; Kleijnen, J.; Moher, D. The PRISMA statement for reporting systematic reviews and meta-analyses of studies that evaluate healthcare interventions: Explanation and elaboration. *BMJ* 2009, 339, b2700. [CrossRef]
- 13. Siebert, T.; Rode, C.; Herzog, W.; Till, O.; Blickhan, R. Nonlinearities make a difference: Comparison of two common Hill-type models with real muscle. *Biol. Cybern.* **2008**, *98*, 133–143. [CrossRef]
- 14. Lei, M.; Wang, Z.; Feng, Z. Detecting nonlinearity of action surface EMG signal. Phys. Lett. A 2001, 290, 297–303. [CrossRef]
- Ho, N.S.K.; Tong, K.Y.; Hu, X.L.; Fung, K.L.; Wei, X.J.; Rong, W.; Susanto, E.A. An EMG-Driven Exoskeleton Hand Robotic Training Device on Chronic Stroke Subjects: Task Training System for Stroke Rehabilitation. In Proceedings of the 2011 IEEE International Conference on Rehabilitation Robotics, Zurich, Switzerland, 29 June–1 July 2011. [CrossRef]
- 16. Gopura, R.; Kiguchi, K. Application of Surface Electromyographic Signals to Control Exoskeleton Robots; IntechOpen: London, UK, 2012. [CrossRef]
- Kiguchi, K.; Hayashi, Y. An EMG-Based Control for an Upper-Limb Power-Assist Exoskeleton Robot. *IEEE Trans. Syst. Man Cybernetics. Part B Cybern.* 2012, 42, 1064–1071. [CrossRef] [PubMed]
- Pang, M.; Guo, S.; Song, Z. Study on the sEMG Driven Upper Limb Exoskeleton Rehabilitation Device in Bilateral Rehabilitation. J. Robot. Mechatron. 2012, 24, 585–594. [CrossRef]
- Delph, M.A.; Fischer, S.A.; Gauthier, P.W.; Luna, C.H.M.; Clancy, E.A.; Fischer, G.S. A Soft Robotic Exomusculature Glove with Integrated sEMG Sensing for Hand Rehabilitation. In Proceedings of the 2013 IEEE 13th International Conference on Rehabilitation Robotics (ICORR), Seattle, WA, USA, 24–26 June 2013. [CrossRef]
- Loconsole, C.; Leonardis, D.; Barsotti, M.; Solazzi, M.; Frisoli, A.; Bergamasco, M.; Troncossi, M.; Foumashi, M.M.; Mazzotti, C.; Castelli, V.P. An emg-based robotic hand exoskeleton for bilateral training of grasp. In Proceedings of the 2013 World Haptics Conference (WHC), Daejeon, Korea, 14–17 April 2013. [CrossRef]
- Su, H.; Li, Z.; Li, G.; Yang, C. EMG-Based Neural Network Control of an Upper-Limb Power-Assist Exoskeleton Robot. In Advances in Neural Networks—ISNN 2013; Hutchison, D., Kanade, T., Kittler, J., Kleinberg, J.M., Mattern, F., Mitchell, J.C., Naor, M., Nierstrasz, O., Pandu Rangan, C., Steffen, B., et al., Eds.; Springer: Berlin/Heidelberg, Germany, 2013; Volume 7952, pp. 204–211. [CrossRef]
- Ngeo, J.; Tamei, T.; Shibata, T.; Orlando, M.F.; Behera, L.; Saxena, A.; Dutta, A. Control of an optimal finger exoskeleton based on continuous joint angle estimation from EMG signals. In Proceedings of the 2013 35th Annual International Conference of the IEEE Engineering in Medicine and Biology Society (EMBC), Osaka, Japan, 3–7 July 2013; pp. 338–341. [CrossRef]
- Ramos, J.L.A.S.; Meggiolaro, M.A. Use of surface electromyography for human amplification using an exoskeleton driven by artificial pneumatic muscles. In Proceedings of the 5th IEEE RAS/EMBS International Conference on Biomedical Robotics and Biomechatronics, Sao Paulo, Brazil, 12–15 August 2014. [CrossRef]
- Kawase, T.; Sakurada, T.; Koike, Y.; Kansaku, K. Estimating joint angles from biological signals for multi-joint exoskeletons. In Proceedings of the 2014 IEEE International Conference on Systems, Man, and Cybernetics (SMC), San Diego, CA, USA, 5–8 October 2014. [CrossRef]
- Loconsole, C.; Dettori, S.; Frisoli, A.; Avizzano, C.A.; Bergamasco, M. An EMG-based approach for on-line predicted torque control in robotic-assisted rehabilitation. In Proceedings of the 2014 IEEE Haptics Symposium (HAPTICS), Houston, TX, USA, 23–26 February 2014; pp. 181–186. [CrossRef]
- Li, Z.; Wang, B.; Sun, F.; Yang, C.; Xie, Q.; Zhang, W. sEMG-based joint force control for an upper-limb power-assist exoskeleton robot. *IEEE J. Biomed. Health Inform.* 2014, 18, 1043–1050. [CrossRef] [PubMed]
- Tang, Z.; Zhang, K.; Sun, S.; Gao, Z.; Zhang, L.; Yang, Z. An Upper-Limb Power-Assist Exoskeleton Using Proportional Myoelectric Control. Sensors 2014, 14, 6677. [CrossRef]
- 28. Kirchner, E.A.; Tabie, M.; Seeland, A. Multimodal movement prediction-towards an individual assistance of patients. *PLoS ONE* **2014**, *9*, e85060. [CrossRef]
- Buongiorno, D.; Barsotti, M.; Sotgiu, E.; Loconsole, C.; Solazzi, M.; Bevilacqua, V.; Frisoli, A. A neuromusculoskeletal model of the human upper limb for a myoelectric exoskeleton control using a reduced number of muscles. In Proceedings of the 2015 IEEE World Haptics Conference (WHC), Evanston, IL, USA, 22–26 June 2015. [CrossRef]
- Riener, R.; Novak, D. Movement Onset Detection and Target Estimation for Robot-Aided Arm Training. *Automatisierungstechnik* 2015, 63, 286–298. [CrossRef]
- Krasin, V.; Gandhi, V.; Yang, Z.; Karamanoglu, M. EMG based elbow joint powered exoskeleton for biceps brachii strength augmentation. In Proceedings of the 2015 International Joint Conference on Neural Networks (IJCNN), Killarney, Ireland, 12–16 July 2015; pp. 1–6. [CrossRef]
- 32. Leonardis, D.; Barsotti, M.; Loconsole, C.; Solazzi, M.; Troncossi, M.; Mazzotti, C.; Castelli, V.P.; Procopio, C.; Lamola, G.; Chisari, C. An EMG-controlled robotic hand exoskeleton for bilateral rehabilitation. *IEEE Trans. Haptics* **2015**, *8*, 140–151. [CrossRef]

- Ullauri, J.B.; Peternel, L.; Ugurlu, B.; Yamada, Y.; Morimoto, J. On the EMG-based torque estimation for humans coupled with a force-controlled elbow exoskeleton. In Proceedings of the 2015 International Conference on Advanced Robotics (ICAR), Istanbul, Turkey, 27–31 July 2015; pp. 302–307. [CrossRef]
- Triwiyanto; Wahyunggoro, O.; Nugroho, H.A.; Herianto. String actuated upper limb exoskeleton based on surface electromyography control. In Proceedings of the 2016 6th International Annual Engineering Seminar (InAES), Yogyakarta, Indonesia, 1–3 August 2016; pp. 176–181. [CrossRef]
- 35. Peternel, L.; Noda, T.; Petrič, T.; Ude, A.; Morimoto, J.; Babič, J. Adaptive Control of Exoskeleton Robots for Periodic Assistive Behaviours Based on EMG Feedback Minimisation. *PLoS ONE* **2016**, *11*, e0148942. [CrossRef]
- Lu, Z.Y.; Chen, X.; Zhang, X.; Tong, K.Y.; Zhou, P. Real-Time Control of an Exoskeleton Hand Robot with Myoelectric Pattern Recognition. *Int. J. Neural Syst.* 2017, 27, 1750009. [CrossRef]
- Li, Z.; Huang, Z.; He, W.; Su, C.Y. Adaptive Impedance Control for an Upper Limb Robotic Exoskeleton Using Biological Signals. IEEE Trans. Ind. Electron. 2017, 64, 1664–1674. [CrossRef]
- Hosseini, M.; Meattini, R.; Palli, G.; Melchiorri, C. A Wearable Robotic Device Based on Twisted String Actuation for Rehabilitation and Assistive Applications. J. Robot. 2017, 2017, 3036468. [CrossRef]
- Mghames, S.; Laghi, M.; Della Santina, C.; Garabini, M.; Catalano, M.; Grioli, G.; Bicchi, A. Design, control and validation of the variable stiffness exoskeleton FLExo. In Proceedings of the 2017 International Conference on Rehabilitation Robotics (ICORR), London, UK, 17–20 July 2017; Volume 2017, pp. 539–546. [CrossRef]
- 40. Hamaya, M.; Matsubara, T.; Noda, T.; Teramae, T.; Morimoto, J. Learning assistive strategies for exoskeleton robots from user-robot physical interaction. *Pattern Recognit. Lett.* **2017**, *99*, 67–76. [CrossRef]
- Yun, Y.; Dancausse, S.; Esmatloo, P.; Serrato, A.; Merring, C.A.; Agarwal, P.; Deshpande, A.D. Maestro: An EMG-driven assistive hand exoskeleton for spinal cord injury patients. In Proceedings of the 2017 IEEE International Conference on Robotics and Automation (ICRA), Singapore, 29 May–3 June 2017; pp. 2904–2910. [CrossRef]
- Irastorza-Landa, N.; Sarasola-Sanz, A.; Lopez-Larraz, E.; Bibian, C.; Shiman, P.; Birbaumer, N.; Ramos-Murguialday, A. Design of continuous EMG classification approaches towards the control of a robotic exoskeleton in reaching movements. In Proceedings of the 2017 International Conference on Rehabilitation Robotics (ICORR), London, UK, 17–20 July 2017; Volume 2017, pp. 128–133. [CrossRef]
- Lambelet, C.; Lyu, M.X.; Woolley, D.; Gassert, R.; Wenderoth, N. The eWrist—A Wearable Wrist Exoskeleton with sEMG-based Force Control for Stroke Rehabilitation. In Proceedings of the 2017 International Conference on Rehabilitation Robotics (ICORR), London, UK, 17–20 July 2017; pp. 726–733. [CrossRef]
- 44. Lince, A.; Celadon, N.; Battezzato, A.; Favetto, A.; Appendino, S.; Ariano, P.; Paleari, M. Design and testing of an under-actuated surface EMG-driven hand exoskeleton. In Proceedings of the 2017 International Conference on Rehabilitation Robotics (ICORR), London, UK, 17–20 July 2017; pp. 670–675. [CrossRef]
- Copaci, D.; Serrano, D.; Moreno, L.; Blanco, D. A High-Level Control Algorithm Based on sEMG Signalling for an Elbow Joint SMA Exoskeleton. *Sensors* 2018, 18, 2522. [CrossRef] [PubMed]
- Zeng, H.; Li, K.; Wei, N.; Song, R.; Tian, X. A semg-controlled robotic hand exoskeleton for rehabilitation in post-stroke individuals. In Proceedings of the 2018 IEEE International Conference on Cyborg and Bionic Systems (CBS), Shenzhen, China, 25–27 October 2018; pp. 652–655. [CrossRef]
- 47. Buongiorno, D.; Barsotti, M.; Barone, F.; Bevilacqua, V.; Frisoli, A. A Linear Approach to Optimize an EMG-Driven Neuromusculoskeletal Model for Movement Intention Detection in Myo-Control: A Case Study on Shoulder and Elbow Joints. *Front. Neurorobot.* **2018**, *12*, 74. [CrossRef]
- Cisnal, A.; Alonso, R.; Turiel, J.P.; Fraile, J.C.; Lobo, V.; Moreno, V. EMG Based Bio-Cooperative Direct Force Control of an Exoskeleton for Hand Rehabilitation: A Preliminary Study. In *Converging Clinical and Engineering Research on Neurorehabilitation III*; Masia, L., Micera, S., Akay, M., Pons, J.L., Eds.; Springer International Publishing: Cham, Switzerland, 2019; Volume 21, pp. 390–394. [CrossRef]
- Jana, M.; Barua, B.G.; Hazarika, S.M. Design and Development of a Finger Exoskeleton for Motor Rehabilitation using Electromyography Signals. In Proceedings of the 2019 23rd International Conference on Mechatronics Technology (ICMT), Salerno, Italy, 23–26 October 2019. [CrossRef]
- 50. Trigili, E.; Grazi, L.; Crea, S.; Accogli, A.; Carpaneto, J.; Micera, S.; Vitiello, N.; Panarese, A. Detection of movement onset using EMG signals for upper-limb exoskeletons in reaching tasks. *J. Neuroeng. Rehabil.* **2019**, *16*, 45. [CrossRef] [PubMed]
- 51. Lei, Z. An upper limb movement estimation from electromyography by using BP neural network. *Biomed. Signal Process. Control* **2019**, *49*, 434–439. [CrossRef]
- 52. Wu, Q.; Chen, B.; Wu, H. Neural-network-enhanced torque estimation control of a soft wearable exoskeleton for elbow assistance. *Mechatronics* **2019**, *63*, 102279. [CrossRef]
- 53. Xiao, F. Proportional myoelectric and compensating control of a cable-conduit mechanism-driven upper limb exoskeleton. *ISA Trans.* **2019**, *89*, 245–255. [CrossRef]
- 54. Lu, Z.; Stampas, A.; Francisco, G.E.; Zhou, P. Offline and online myoelectric pattern recognition analysis and real-time control of a robotic hand after spinal cord injury. *J. Neural Eng.* **2019**, *16*, 036018. [CrossRef] [PubMed]
- 55. Secciani, N.; Bianchi, M.; Meli, E.; Volpe, Y.; Ridolfi, A. A novel application of a surface ElectroMyoGraphy-based control strategy for a hand exoskeleton system: A single-case study. *Int. J. Adv. Robot. Syst.* **2019**, *16*, 1. [CrossRef]

- Burns, M.K.; Pei, D.; Vinjamuri, R. Myoelectric control of a soft hand exoskeleton using kinematic synergies. *IEEE Trans. Biomed. Circuits Syst.* 2019, 13, 1351–1361. [CrossRef] [PubMed]
- 57. Lu, L.; Wu, Q.; Chen, X.; Shao, Z.; Chen, B.; Wu, H. Development of a sEMG-based torque estimation control strategy for a soft elbow exoskeleton. *Robot. Auton. Syst.* **2019**, *111*, 88–98. [CrossRef]
- 58. Li, X.; Liu, S.; Chang, Y.; Li, S.; Fan, Y.; Yu, H. A Human Joint Torque Estimation Method for Elbow Exoskeleton Control. *Int. J. Humanoid Robot.* **2020**, *17*, 03. [CrossRef]
- Lotti, N.; Xiloyannis, M.; Durandau, G.; Galofaro, E.; Sanguineti, V.; Masia, L.; Sartori, M. Adaptive Model-Based Myoelectric Control for a Soft Wearable Arm Exosuit: A New Generation of Wearable Robot Control. *IEEE Robot. Autom. Mag.* 2020, 27, 43–53. [CrossRef]
- 60. Hosseini, M.; Meattini, R.; San-Millan, A.; Palli, G.; Melchiorri, C.; Paik, J. A sEMG-Driven Soft ExoSuit Based on Twisted String Actuators for Elbow Assistive Applications. *IEEE Robot. Autom. Lett.* **2020**, *5*, 4094–4101. [CrossRef]
- 61. Da Silva, L.D.L.; Pereira, T.F.; Leithardt, V.R.Q.; Seman, L.O.; Zeferino, C.A. Hybrid Impedance-Admittance Control for Upper Limb Exoskeleton Using Electromyography. *Appl. Sci.* **2020**, *10*, 7146. [CrossRef]
- Treussart, B.; Geffard, F.; Vignais, N.; Marin, F. Controlling an upper-limb exoskeleton by EMG signal while carrying unknown load. In Proceedings of the 2020 IEEE International Conference on Robotics and Automation (ICRA), Paris, France, 31 May–31 August 2020; pp. 9107–9113. [CrossRef]
- 63. Liu, H.; Tao, J.; Lyu, P.; Tian, F. Human-robot cooperative control based on sEMG for the upper limb exoskeleton robot. *Robot. Auton. Syst.* **2020**, *125*, 103350. [CrossRef]
- McDonald, C.G.; Sullivan, J.L.; Dennis, T.A.; O'Malley, M.K. A Myoelectric Control Interface for Upper-Limb Robotic Rehabilitation Following Spinal Cord Injury. *IEEE Trans. Neural Syst. Rehabil. Eng.* 2020, 28, 978–987. [CrossRef]
- 65. Castiblanco, J.C.; Mondragon, I.F.; Alvarado-Rojas, C.; Colorado, J.D. Assist-As-Needed Exoskeleton for Hand Joint Rehabilitation Based on Muscle Effort Detection. *Sensors*, **2021**, *21*, 4372. [CrossRef] [PubMed]
- 66. Liu, Y.; Li, X.; Zhu, A.; Zheng, Z.; Zhu, H. Design and evaluation of a surface electromyography-controlled lightweight upper arm exoskeleton rehabilitation robot. *Int. J. Adv. Robot. Syst.* **2021**, *18*, 3. [CrossRef]
- 67. Yang, Z.; Guo, S.; Liu, Y.; Hirata, H.; Tamiya, T. An intention-based online bilateral training system for upper limb motor rehabilitation. *Microsyst. Technol.* 2021, 27, 211–222. [CrossRef]
- 68. Zhou, Y.; Chen, C.; Alshahrani, Y.; Cheng, M.; Xu, G.; Li, M.; Zhou, W.; Wu, L.; Kakos, B.; Frush, T.; et al. Real-time Multiple-Channel Shoulder EMG Processing for a Rehabilitative Upper-limb Exoskeleton Motion Control Using ANN Machine Learning. In Proceedings of the 2021 27th International Conference on Mechatronics and Machine Vision in Practice (M2VIP), Shanghai, China, 26–28 November 2021. [CrossRef]
- Cisnal, A.; Pérez-Turiel, J.; Fraile, J.C.; Sierra, D.; Fuente, E.d.l. RobHand: A Hand Exoskeleton With Real-Time EMG-Driven Embedded Control. Quantifying Hand Gesture Recognition Delays for Bilateral Rehabilitation. *IEEE Access* 2021, 9, 137809–137823. [CrossRef]
- Xiao, F.; Gu, L.; Ma, W.; Zhu, Y.; Zhang, Z.; Wang, Y. Real time motion intention recognition method with limited number of surface electromyography sensors for A 7-DOF hand/wrist rehabilitation exoskeleton. *Mechatronics* 2021, 79, 102642. [CrossRef]
- 71. Liu, C.; Liang, H.; Murata, Y.; Li, P.; Ueda, N.; Matsuzawa, R.; Zhu, C. A wearable lightweight exoskeleton with full degrees of freedom for upper-limb power assistance. *Adv. Robot.* **2021**, *35*, 413–424. [CrossRef]
- 72. Xie, C.; Yang, Q.; Huang, Y.; Su, S.W.; Xu, T.; Song, R. A Hybrid Arm-Hand Rehabilitation Robot With EMG-Based Admittance Controller. *IEEE Trans. Biomed. Circuits Syst.* 2021, *15*, 1332–1342. [CrossRef]
- 73. van Eck, N.J.; Waltman, L. Software survey: VOSviewer, a computer program for bibliometric mapping. *Scientometrics* **2010**, *84*, 523–538. [CrossRef]
- Lenzi, T.; Rossi, S.M.M.; Vitiello, N.; Carrozza, M.C. Proportional EMG control for upper-limb powered exoskeletons. In Proceedings of the 2011 Annual International Conference of the IEEE Engineering in Medicine and Biology Society, Boston, MA, USA, 30 August–3 September 2011; Volume 2011, pp. 628–631. [CrossRef]
- 75. Brown, S.H.; McGill, S. Revising the EMG-torque relationship of the trunk musculature: Effects of antagonistic co-contraction. *Am. Soc. Biomech.* **1997**, *48*, 411–426.
- 76. Rosen, J.; Fuchs, M.B.; Arcan, M. Performances of hill-type and neural network muscle models-toward a myosignal-based exoskeleton. *Comput. Biomed. Res.* **1999**, *32*, 415–439. [CrossRef]
- Rosen, J.; Brand, M.; Fuchs, M.B.; Arcan, M. A myosignal-based powered exoskeleton system. *IEEE Trans. Syst. Man Cybern. Part A Syst. Humans* 2001, 31, 210–222. [CrossRef]
- Perreault, E.J.; Heckman, C.J.; Sandercock, T.G. Hill muscle model errors during movement are greatest within the physiologically relevant range of motor unit firing rates. J. Biomech. 2003, 36, 211–218. [CrossRef]
- 79. Masuda, K.; Masuda, T.; Sadoyama, T.; Mitsuharu, I.; Katsuta, S. Changes in surface EMG parameters during static and dynamic fatiguing contractions. *J. Electromyogr. Kinesiol.* **1999**, *9*, 39–46. [CrossRef]
- Laursen, B.; Jensen, B.R.; Németh, G.; Sjøgaard, G. A model predicting individual shoulder muscle forces based on relationship between electromyographic and 3D external forces in static position. *J. Biomech.* 1998, *31*, 731–739. [CrossRef]
- 81. Cavallaro, E.E.; Rosen, J.; Perry, J.C.; Burns, S. Real-Time Myoprocessors for a Neural Controlled Powered Exoskeleton Arm. *IEEE Trans. Biomed. Eng.* **2006**, *53*, 2387–2396. [CrossRef]

- 82. Serrancolí, G.; Kinney, A.L.; Fregly, B.J.; Font-Llagunes, J.M. Neuromusculoskeletal Model Calibration Significantly Affects Predicted Knee Contact Forces for Walking. *J. Biomech. Eng.* **2016**, *138*, 081001. [CrossRef]
- 83. Phinyomark, A.; Phukpattaranont, P.; Limsakul, C. Feature reduction and selection for EMG signal classification. *Expert Syst. Appl.* **2012**, *39*, 7420–7431. [CrossRef]
- Arteaga, M.V.; Castiblanco, J.C.; Mondragon, I.F.; Colorado, J.D.; Alvarado-Rojas, C. EMG-based adaptive trajectory generation for an exoskeleton model during hand rehabilitation exercises. In Proceedings of the 2020 8th IEEE RAS/EMBS International Conference for Biomedical Robotics and Biomechatronics (BioRob), New York, NY, USA, 29 November–1 December 2020; pp. 416–421. [CrossRef]
- 85. Kaelbling, L.P.; Littman, M.L.; Moore, A.W. Reinforcement Learning: A Survey. J. Artif. Intell. Res. 1996, 4, 237–285. [CrossRef]
- 86. Pilarski, P.M.; Dawson, M.R.; Degris, T.; Fahimi, F.; Carey, J.P.; Sutton, R.S. Online human training of a myoelectric prosthesis controller via actor-critic reinforcement learning. *IEEE Int. Conf. Rehabil. Robot.* **2011**, 2011, 5975338. [CrossRef]
- 87. Luo, S.; Androwis, G.; Adamovich, S.; Su, H.; Nunez, E.; Zhou, X. Reinforcement Learning and Control of a Lower Extremity Exoskeleton for Squat Assistance. *Front. Robot. AI* **2021**, *8*. [CrossRef]
- 88. Wu, W.; Saul, K.R.; Huang, H. Using Reinforcement Learning to Estimate Human Joint Moments From Electromyography or Joint Kinematics: An Alternative Solution to Musculoskeletal-Based Biomechanics. J. Biomech. Eng. 2021, 143. [CrossRef] [PubMed]
- 89. Deisenroth, M.P.; Rasmussen, C.E. PILCO: A Model-Based and Data-Efficient Approach to Policy Search. In Proceedings of the 28th International Conference on Machine Learning, Bellevue, WA, USA, 28 June–2 July 2011. [CrossRef]
- Jaber, H.A.; Rashid, M.T.; Fortuna, L. Online myoelectric pattern recognition based on hybrid spatial features. *Biomed. Signal Process. Control* 2021, 66, 102482. [CrossRef]
- 91. José, M.N.V.; Dias, F.M.; Alexandre Manuel, M. Neuro-Fuzzy Systems: A Survey. Wseas Trans. Syst. Arch. 2004, 3, 1–6.
- 92. Kiguchi, K.; Esaki, R.; Fukuda, T. Development of a wearable exoskeleton for daily forearm motion assist. *Adv. Robot.* 2005, 19, 751–771. [CrossRef]
- 93. Kiguchi, K.; Tanaka, T.; Fukuda, T. Neuro-fuzzy control of a robotic exoskeleton with EMG signals. *IEEE Trans. Fuzzy Syst.* 2004, 12, 481–490. [CrossRef]
- 94. Phinyomark, A.; Quaine, F.; Charbonnier, S.; Serviere, C.; Tarpin-Bernard, F.; Laurillau, Y. EMG feature evaluation for improving myoelectric pattern recognition robustness. *Expert Syst. Appl.* **2013**, *40*, 4832–4840. [CrossRef]
- 95. Geng, Y.J.; Samuel, O.W.; Wei, Y.; Li, G.L. Improving the Robustness of Real-Time Myoelectric Pattern Recognition against Arm Position Changes in Transradial Amputees. *Biomed Res. Int.* 2017, 2017, 5090454. [CrossRef]
- 96. Tkach, D.; Huang, H.; Kuiken, T.A. Study of stability of time-domain features for electromyographic pattern recognition. *J. NeuroEng. Rehabil.* **2010**, *7*, 21. [CrossRef]
- Boschmann, A.; Kaufmann, P.; Platzner, M.; Winkler, M. Towards multi-movement hand prostheses: Combining adaptive classification with high precision sockets. In Proceedings of the 2nd European Conference Technically Assisted Rehabilitation, Berlin, Germany, 18–19 March 2009.
- 98. Scheme, E.; Englehart, K. Electromyogram pattern recognition for control of powered upper-limb prostheses: State of the art and challenges for clinical use. *J. Rehabil. Res. Dev.* **2011**, *48*, 643. [CrossRef]
- 99. He, J.; Zhang, D.; Sheng, X.; Li, S.; Zhu, X. Invariant Surface EMG Feature Against Varying Contraction Level for Myoelectric Control Based on Muscle Coordination. *IEEE J. Biomed. Health Inform.* **2015**, *19*, 874–882. [CrossRef]
- Zhang, X.; Li, X.; Samuel, O.W.; Huang, Z.; Fang, P.; Li, G. Improving the robustness of electromyogram-pattern recognition for prosthetic control by a postprocessing strategy. *Front. Neurorobot.* 2017, *11*, 51. [CrossRef] [PubMed]
- Young, A.J.; Hargrove, L.J.; Kuiken, T.A. Improving Myoelectric Pattern Recognition Robustness to Electrode Shift by Changing Interelectrode Distance and Electrode Configuration. *IEEE Trans. Biomed. Eng.* 2012, 59, 645–652. [CrossRef] [PubMed]
- 102. Asghari Oskoei, M.; Hu, H. Myoelectric control systems: A survey. Biomed. Signal Process. Control 2007, 2, 275–294. [CrossRef]
- 103. Kaufmann, P.; Englehart, K.; Platzner, M. Fluctuating emg signals: Investigating long-term effects of pattern matching algorithms. In Proceedings of the 2010 Annual International Conference of the IEEE Engineering in Medicine and Biology, Buenos Aires, Argentina, 31 August–4 September 2010; pp. 6357–6360. [CrossRef]
- 104. Novak, D.; Riener, R. A survey of sensor fusion methods in wearable robotics. Robot. Auton. Syst. 2015, 73, 155–170. [CrossRef]
- 105. Turvey, M.T. Action and perception at the level of synergies. Hum. Mov. Sci. 2007, 26, 657-697. [CrossRef]
- 106. Mesin, L. Crosstalk in surface electromyogram: Literature review and some insights. *Phys. Eng. Sci. Med.* **2020**, *43*, 481–492. [CrossRef]
- 107. Kang, H.B.; Wang, J.H. Adaptive control of 5 DOF upper-limb exoskeleton robot with improved safety. *ISA Trans.* 2013, 52, 844–852. [CrossRef]
- López, N.M.; di Sciascio, F.; Soria, C.M.; Valentinuzzi, M.E. Robust EMG sensing system based on data fusion for myoelectric control of a robotic arm. *Biomed. Eng. Online* 2009, *8*, 1–13. [CrossRef]
- Bao, T.; Xie, S.Q.; Yang, P.; Zhou, P.; Zhang, Z.Q. Toward Robust, Adaptive and Reliable Upper-Limb Motion Estimation Using Machine Learning and Deep Learning–A Survey in Myoelectric Control. *IEEE J. Biomed. Health Inform.* 2022, 26, 3822–3835. [CrossRef]
- Englehart, K.; Hudgins, B. A robust, real-time control scheme for multifunction myoelectric control. *IEEE Trans. Biomed. Eng.* 2003, 50, 848–854. [CrossRef]

- Bao, T.; Zaidi, S.A.R.; Xie, S.Q.; Yang, P.; Zhang, Z.Q. CNN Confidence Estimation for Rejection-Based Hand Gesture Classification in Myoelectric Control. *IEEE Trans. Hum. Mach. Syst.* 2021, 52, 99–109. [CrossRef]
- Edwards, A.L.; Dawson, M.R.; Hebert, J.S.; Sherstan, C.; Sutton, R.S.; Chan, K.M.; Pilarski, P.M. Application of real-time machine learning to myoelectric prosthesis control: A case series in adaptive switching. *Prosthetics Orthot. Int.* 2016, 40, 573–581. [CrossRef] [PubMed]
- 113. Li, G.; Schultz, A.E.; Kuiken, T.A. Quantifying pattern recognition—Based myoelectric control of multifunctional transradial prostheses. *IEEE Trans. Neural Syst. Rehabil. Eng.* 2010, *18*, 185–192. [CrossRef] [PubMed]
- 114. Meattini, R.; Chiaravalli, D.; Palli, G.; Melchiorri, C. sEMG-Based Human-in-the-Loop Control of Elbow Assistive Robots for Physical Tasks and Muscle Strength Training. *IEEE Robot. Autom. Lett.* **2020**, *5*, 5795–5802. [CrossRef]